Author's response to reviews

Title: Lessons from the implementation of LLIN distribution campaign in Kwara State, North Central Nigeria

Authors:

Abiodun Obembe (abiodunobembe@yahoo.com)
Okorie O Anyaele (oo.anyaele@mail.ui.edu.ng)
Adedayo O Oduola (dayooduola@yahoo.co.uk)

Version: 2 Date: 7 February 2014

Author's response to reviews:

Dear Editor,

Thanks for your observation.

The methodology section has now been divided into different sub-sections including a part describing ethical consideration.

Thanks
Abiodun OBEMBE
Corresponding author.